Author's response to reviews

Title: Paediatric Investigators Collaborative Network on Infections in Canada (PICNIC) study of aseptic meningitis

Authors:

Bonita E Lee (bonitlee@ualberta.ca)
Rupesh Chawla (rupesh.chawla@calgaryhealthregion.ca)
Joanne M Langley (joanne.langley@dal.ca)
Sarah E Forgie (sarahforgie@cha.ab.ca)
Mohammed Al-Hosni (malhosni@hotmail.com)
Krista Baerg (kribae@sasktel.net)
Enteresar Husain (entesar@qualitynet.net)
James Strong (Jim_Strong@phac-aspc.gc.ca)
Joan L Robinson (jr3@ualberta.ca)
Upton Allen (upton.allen@sickkids.ca)
Barbara J Law (viktorbj@shaw.ca)
Simon Dobson (sdobson@cw.bc.ca)
Dele H Davies (daviesde@msu.edu)

Version: 4 Date: 3 April 2006

Author’s response to reviews:

Dear Editor,

I have corrected the list of authors to correspond exactly to those in the manuscript. However, I was not able to make changes in the title as "Paediatric Investigators Collaborative Network on Infections in Canada" is the name of a research group so the words are capitalized. Please contact me if you require further changes. I apologize for the delay as I've been on vacation but am back now so will try to answer the same day if further changes are needed.

Yours sincerely,

Joan Robinson